Author’s response to reviews

Title: The Acute Effects of Alcohol on Auditory Thresholds

Authors:

T Upile (mrtupile@yahoo.com)
P Andrews (Peterandrews1625@sky.com)
W Jerjes (waseem_wk1@yahoo.co.uk)
S Singh (s154singh@yahoo.com)
F Sipaul (Fabian@tiscali.com)
Sar Nouraei (Sarnouraei@hotmail.com)
M El Maaytah (elmaaytah@hotmail.com)
C Hopper (c.hopper@ucl.ac.uk)
J Graham (James14357@yahoo.com)
A Wright (anthonywright@yahoo.co.uk)

Version: 3 Date: 28 July 2007

Author’s response to reviews:

Dear reviewer 1

Many thanks for your very diligent review. Please note that we found it to be very constructive. We had engaged a Professor of Audiology previously however our error was to transpose the scale which was on the left was to the right unfortunately the graph placeholder - was also transposed making e.g. 10dB wrongly -10dB. We thank you again. All measurements were done by a full time qualified audiologist. Please note that figure one is not an audiogram even though it follows the general plot rules you have outlined but represents the mean hearing thresholds over the selected frequencies. We have added the error bars as requested.

Many Thanks

The authors